Author’s response to reviews

Title: Implementing Alcohol Misuse SBIRT in a National Cohort of Pediatric Trauma Centers - a type III hybrid effectiveness-implementation trial

Authors:

Michael Mello (michael_mello_md@brown.edu)
Sara Becker (sara_becker@brown.edu)
Julie Bromberg (jbromberg@lifespan.org)
Janette Baird (jbaird@lifespan.org)
Mark Zonfrillo (zonfrillo@brown.edu)
Anthony Spirito (anthony_spirito@brown.edu)

Version: 1 Date: 01 Feb 2018

Author’s response to reviews:

No specific reviewer/editor comments were provided as of yet.